

Author's response to reviews

Title: Early coordinated rehabilitation in acute phase after hip fracture - a model for increased patient participation.

Authors:

Gillian Asplin (gillian.asplin@vgregion.se)

Gunnel Carlsson (gunnel.carlsson@neuro.gu.se)

Lena Ziden (lena.ziden@vgregion.se)

Gunilla Kjellby Wendt (gunilla.wendt@telia.com)

Version: 1 **Date:** 07 Sep 2017

Author's response to reviews:

The authors' response letter has been included as a supplementary file